Author’s response to reviews

Title: Consultant medical trainers, modernising medical careers (MMC) and the European time directive (EWTD): tensions and challenges in a changing medical education context.

Authors:

Maria Tsouroufli (M.Tsouroufli@uea.ac.uk)
Heather Payne (payneeh@cf.ac.uk)

Version: 3 Date: 23 April 2008

Author’s response to reviews:

I have incorporated the minor concerns of Referee 2. I have also made all the formatting changes requested by BMC Medical Education so that the manuscript confirms to the standards of the journal. A statement on ethical approval for the study has been included in the Methods section as well as the name of the body which gave approval and a reference number.